Author's response to reviews

Title: Vagal baroreflex activation resulting in acute coronary stent thrombus associated with myocardial infarction: a case report

Authors:

Yan-Yan Jing

Version: 4 Date: 22 August 2014

Author's response to reviews:

Dear Editor and Reviewers,

Following your instruction and comments from the reviewers, we have carefully re-revised manuscript. All revisions are highlighted through the text.

In addition, some words of the manuscript were reduced, but it cannot be reduced further or it will damage manuscript's information.

We wish these revisions are good for your acceptance of the manuscript.

Thanks and warm regards,

Liu-Dong Li, MD.

Department of Cardiology,
Yuhuangding Hospital
Qingdao Medical College, Qingdao University
Yantai, Shandong Province
China